Author's response to reviews

Title: Repeatability and Variability of Region-of-Interest Methods Using Quantitative Diffusion Tensor MR Imaging of the Brain

Authors:

Ullamari K Hakulinen (ullamari.hakulinen@gmail.com)
Antti Brander (antti.brander@pshp.fi)
Pertti Ryymin (pertti.ryymin@pshp.fi)
Juha Öhman (juha.ohman@pshp.fi)
Seppo Soimakallio (seppo.soimakallio@pshp.fi)
Prasun Dastidar (prasun.dastidar@pshp.fi)
Hannu Eskola (hannu.eskola@tut.fi)

Version: 2 Date: 18 April 2012

Author's response to reviews:

Dear Editors,

I have corrected the manuscript in accordance as you suggest. The largest changes are located within the discussion. Repairs also include a cite to our previous studies which have been published in IFMBE Proceedings 2011.

Yours sincerely,
Ullamari Hakulinen